 Vc/5 april 05
Week 14  starting 4 april 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  4 April 05 @ 4/4/2005 2:44:02 PM   
	7
	A1
	MBBL1193
	PT 10 Min. NRJ 11850 A
	2.35 HV

	6
	A2
	MBBR3307
	Prep 0 Scan Bench
	295.89

	12
	B1
	SSS104
	Prep 3 CDW1 HV
	294.28

	11
	B2
	MBBL1211
	PT 12 Warm Up
	61.61

	4
	C1
	MBBL1212
	Prep 5 Pumping + Cool Down
	106.03

	3
	C2
	MBAR3254
	PT 12 Warm Up
	26.27

	2
	D1
	SSS117
	heater delay - 1/2 HF
	1.92

	8
	D2
	SSS022
	PT 3 QH Measure
	2.35

	5
	E1
	MBBR3315
	Prep 2.1 IAP @ warm LF
	296.52

	9
	E2
	MBAL1210
	Prep 5 Pumping + Cool Down
	1.90

	10
	F1
	MBAL2136
	Prep 5 Pumping + Cool Down
	201.39

	1
	F2
	MBBR3267
	PT 12 Warm Up
	286.25


Temperature data are taken from TBSM18 site
Coordinator this week: Pierre; MTM support on call :   16-3602 .

14 Magnets Tested  ( not counting 1214)
11  dipoles( 1 with MM)+ 3  SSS(0  with MM) +  0  SSSS (with MM)
    Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :
General Points 
…
New AQA
> > 
> > 
> > -------- Original Message -------- Subject:  Improvements in 
> > software and times for operation friendly systems 
> > Date:  Tue, 05 Apr 2005 13:40:35 +0200 
> > From:  chohan <vinod.chohan@cern.ch> 
> > To:  Louis Walckiers <Louis.Walckiers@cern.ch> 
> > 
> > 
> > Quelques modification seront envisageable si nous avons plus de 
> > flexibilite sur les application SM18 : 
> > 
> > *Pour TEMA : 
> > -Le message ("Are you sure that the magnet is not a repeated 
> > magnet") arrivant sur le TEMA lors du lancement de chaque test. 
> > Dois etre place que pour les "IAP all HF" et "IAP at cold" dans le 
> > cas d'un cycle thermique de l'aimant. 
> > -Un message lors du changement de banc actif entre SSS a Dipole 
> > devrait avertir les operateurs a verifier "quench heater connetion" 
> > et "Henry's setting power converter", sur cluster B seulement. 
> > -La possibilite de switcher OFF Hardware Recognition. 
> > 
> > 
> > *Pour AQA v.6 : 
> > -La selection du pens en analyse manuel devrait etre automatique sur 
> > l'application HFoff . 
> > -La selection de "scale global" et les curseurs "free" par defaut a 
> > l'ouverture de l'application. 
> > -Une valeur affiche dans le MIITS manuel pour le delta T entre 
> > l'origine du quench et la detection potaim card (200 mv). Demander 
> > depuis debut decembre. 
> > -Certaine valeurs devraient etre plus lisible (MIITS manuel). 
> > 
> > *Pour IAP sur SSS : 
> > -La lecture des "voltage taps" devrait se faire a partir d'une 
> > application IAPSSS, et non part lfoff, ce qui a chaque fois 
> > complique le travail pour les operateurs lorsque l'on passe de 
> > dipole a SSS 
> > 
> >  
  AOB

Back to Homepage
	Magnets under Test from 27-03-2005 to 02-04-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL1196 
	[image: image1.png]



	


	21-03-2005 
	34 h.
	31 h.
	85 h.
	16 h.
	17 h.
	29-03-2005 
	183 h.
	
	

	2
	TBB1
	SSS125 
	[image: image3.png]



	


	25-03-2005 
	33 h.
	23 h.
	19 h.
	10 h.
	22 h.
	30-03-2005 
	107 h.
	
	

	3
	TBB2
	MBBL2157 
	[image: image5.png]



	


	26-03-2005 
	34 h.
	30 h.
	28 h.
	17 h.
	12 h.
	31-03-2005 
	121 h.
	
	

	4
	TBC1
	MBAL1200 
	[image: image7.png]



	


	28-03-2005 
	18 h.
	24 h.
	22 h.
	14 h.
	8 h.
	01-04-2005 
	86 h.
	
	

	5
	TBC1
	MBAL2155 
	[image: image9.png]



	


	21-03-2005 
	31 h.
	30 h.
	26 h.
	14 h.
	34 h.
	28-03-2005 
	135 h.
	
	

	6
	TBC1
	MBBL1214 
	[image: image11.png]



	


	01-04-2005 
	25 h.
	-25 h.
	27 h.
	1 h.
	4 h.
	02-04-2005 
	32 h.
	
	

	7
	TBC2
	MBAL2125 
	[image: image13.png]



	


	25-03-2005 
	32 h.
	29 h.
	14 h.
	14 h.
	17 h.
	30-03-2005 
	106 h.
	
	

	8
	TBD1
	SSS028 
	[image: image15.png]



	


	23-03-2005 
	22 h.
	20 h.
	17 h.
	9 h.
	77 h.
	29-03-2005 
	145 h.
	
	

	9
	TBD2
	SSS025 
	[image: image17.png]



	


	25-03-2005 
	40 h.
	14 h.
	-
	-
	-
	31-03-2005 
	118 h.
	
	

	10
	TBE1
	MBBR3294 
	[image: image19.png]



	


	19-03-2005 
	26 h.
	97 h.
	20 h.
	11 h.
	35 h.
	29-03-2005 
	189 h.
	
	

	11
	TBE2
	MBAL3080 
	[image: image21.png]



	


	29-03-2005 
	26 h.
	-19 h.
	58 h.
	13 h.
	8 h.
	02-04-2005 
	86 h.
	
	

	12
	TBE2
	MBBR3312 
	[image: image23.png]



	


	22-03-2005 
	27 h.
	32 h.
	31 h.
	15 h.
	15 h.
	29-03-2005 
	120 h.
	
	

	13
	TBF1
	MBAR3299 
	[image: image25.png]



	


	24-03-2005 
	37 h.
	22 h.
	30 h.
	15 h.
	12 h.
	29-03-2005 
	116 h.
	
	

	14
	TBF1
	MBAR3304 
	[image: image27.png]



	


	29-03-2005 
	27 h.
	30 h.
	14 h.
	13 h.
	10 h.
	02-04-2005 
	94 h.
	
	

	15
	TBF2
	MBBL2162 
	[image: image29.png]



	


	25-03-2005 
	50 h.
	24 h.
	17 h.
	15 h.
	17 h.
	30-03-2005 
	123 h.
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL1196
	12389 A.
(D1_L)
	12382 A.
(D1_L_S04)
	12263 A.
(D1_L)
	12499 A.
(D1_U_MRB)
	12708 A.
(D1_L_MRB)
	0 A.
(No Data)
	12772 A.
(D1-L)
	12846 A.
(D1_L)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-

	2
	SSS125
	11814 A.
(QI-34)
	12285.9 A.
(QE-12)
	[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL2157
	11573 A.
(D1_U)
	11959 A.
(D2_U)
	12462.3 A.
(D2 - U)
	12758.7 A.
(D1-L)
	[image: image33.png]



	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL1200
	11989.3 A.
(D1-U)
	12558 A.
(D2_U)
	[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL2155
	11255 A.
(D1-L)
	12675 A.
(D2-L)
	[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL1214
	[image: image36.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAL2125
	11913 A.
(D2_U)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS028
	11634 A.
(QI - 12)
	12227 A.
(QE - 34)
	[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS025
	11702.8 A.
(QI_34)
	11759 A.
(QE_12)
	12202 A.
(QE-12)
	12397 A.
(QE-12)
	12550 A.
(QI_34)
	12643 A.
(QE_34)
	12773.1 A.
(QI_12)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-

	10
	MBBR3294
	11917.4 A.
(D2_L)
	12302.4 A.
(D1_L)
	[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL3080
	10540.7 A.
(D2_L)
	12197 A.
(D1_L)
	[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBR3312
	8477 A.
(D2 - L)
	12164 A.
(D2 - U)
	12441.5 A.
(D2-U)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR3299
	10237 A.
(D2_U)
	11116 A.
(D1_U)
	12159 A.
(D2_L)
	12690 A.
(D2_U)
	12858 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAR3304
	12281 A.
(D1-U)
	12581 A.
(D2-U)
	[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBL2162
	11222 A.
(D2_U)
	12465.5 A.
(D1-L)
	[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


 
Week 13 march 27  to 2 April  05
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	From 27-03-2005 to 02-04-2005 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBBL2162 
	False 
	25-03-2005 
	50 h.
	24 h.
	17 h.
	15 h.
	17 h.
	30-03-2005 
	123 h.
	2 

	2
	TBF1
	MBAR3299 
	False 
	24-03-2005 
	37 h.
	22 h.
	30 h.
	15 h.
	12 h.
	29-03-2005 
	116 h.
	4 

	3
	TBF1
	MBAR3304 
	False 
	29-03-2005 
	27 h.
	30 h.
	14 h.
	13 h.
	10 h.
	02-04-2005 
	94 h.
	2 

	4
	TBE2
	MBAL3080 
	False 
	29-03-2005 
	26 h.
	-19 h.
	58 h.
	13 h.
	8 h.
	02-04-2005 
	86 h.
	2 

	5
	TBE2
	MBBR3312 
	False 
	22-03-2005 
	27 h.
	32 h.
	31 h.
	15 h.
	15 h.
	29-03-2005 
	120 h.
	3 

	6
	TBE1
	MBBR3294 
	False 
	19-03-2005 
	26 h.
	97 h.
	20 h.
	11 h.
	35 h.
	29-03-2005 
	189 h.
	2 

	7
	TBD2
	SSS025 
	False 
	25-03-2005 
	40 h.
	14 h.
	199 h.
	0 h.
	-135 h.
	31-03-2005 
	118 h.
	7 

	8
	TBD1
	SSS028 
	False 
	23-03-2005 
	22 h.
	20 h.
	17 h.
	9 h.
	77 h.
	29-03-2005 
	145 h.
	2 

	9
	TBC2
	MBAL2125 
	False 
	25-03-2005 
	32 h.
	29 h.
	14 h.
	14 h.
	17 h.
	30-03-2005 
	106 h.
	1 

	10
	TBC1
	MBAL1200 
	False 
	28-03-2005 
	18 h.
	24 h.
	22 h.
	14 h.
	8 h.
	01-04-2005 
	86 h.
	2 

	11
	TBC1
	MBAL2155 
	False 
	21-03-2005 
	31 h.
	30 h.
	26 h.
	14 h.
	34 h.
	28-03-2005 
	135 h.
	2 

	12
	TBC1
	MBBL1214 
	False 
	01-04-2005 
	25 h.
	-25 h.
	27 h.
	1 h.
	4 h.
	02-04-2005 
	32 h.
	0 

	13
	TBB2
	MBBL2157 
	False 
	26-03-2005 
	34 h.
	30 h.
	28 h.
	17 h.
	12 h.
	31-03-2005 
	121 h.
	4 

	14
	TBB1
	SSS125 
	False 
	25-03-2005 
	33 h.
	23 h.
	19 h.
	10 h.
	22 h.
	30-03-2005 
	107 h.
	2 

	15
	TBA1
	MBBL1196 
	True 
	21-03-2005 
	34 h.
	31 h.
	85 h.
	16 h.
	17 h.
	29-03-2005 
	183 h.
	8 
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Dipoles ARCSSSS
Period
Fresh Repeat Total MM % Repeat % MM Fresh Repeat Total MM % Repeat % MM [Fresh|

Year 2002-2003 95
Year 2004 356

Year 2006

Year 2007

C Statistics

[Total Fresh Magnets Tested Total Magnets to be Tested (~)
(1706 11 % Repests)

[Total Repeat Magnets Tested

[Total Magnets Tests Balance Magnets (~)
— Tests Rate per month

required 62.3 |
|Total % Retested 9.50 to complete by Dec 2006





	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _29_A 

Tuesday _29_N


	TBA2
	File C0503291927.a07 & The Quench File  C0503300001.ka01 (for extended program test 1: Minimum energy at 6KA) not transferred to storage.
	AB/CO
	TEMA
	Analysed  in dacq-disk2

	Tuesday _29_N

	TBA2
	SPA occurred during the cleansing cycle of the LHC-Cycle-long-inj-10000s test. 
	ACR
	
	Test re-launched

	Wednesday_30_M

	TBA2 
	Shaft 23, sector s0001, coil A: broken into the TRU
	MTM
	
	Checked By Maryline. Continue magnetic measurements with E coil.

	Wednesday_30_A

	ALL
	Problem with siemens MOBIC T8: when starting it up it is asking for a USB driver name. 
All the network parameters were lost.  
	 
	
	Richard advised to make a reset. 
Richard advised toset the network parameters by re-starting the home page of internet explorer to:  https://edms.cern.ch/asbuilt/plsql/mtf.home . 

	Wednesday_30_N

	TBA2
	AQA 6.0 could not run for the file C0503302304.ka01, due to the error - "MPdata: no datafile". The old version of AQA ran OK.
	AB/CO
	AQA
	 

	Wednesday_30_N

	TBA2
	Quench in Vbus-ext during the pre-cycle of LHC cycle_PF_1800s. Cryo reported a temperature increase in the current leads.
Restarted LHC Cycle, but cryo noticed that the Current Lead temperature was rising. Trying to increase the cooling resulted in a loss of Cryo OK.
	ACR
	
	TC and N.Sammut decided to re-start a cycle with along plateau at nominal. ACR team will profit to check working of the interlock system.

	Thursday _31_M


	TBD2
	TBD2: wrong label name for SSS022(D instead of F according to MTF) non conformity must be created.
	
	
	TC called N.Bourcey. He confirmed that the SSS is a Focussing.NC opened.

	Thursday _31_A

	TBE2
	MB3080: PT5.2 showed an AQA alert on QH Tau Jmp V < -1.0. Informed Richard & TC.
	MTM
	AQA
	TC and Richard checked the result and told to continue with PTE 3KA test after discussion with Pierre. Parasitic signal probably created the alarm

	Thursday _31_A

	TBE2
	MB3080 Test : Training Quench.

Description : Some jump in Trigger and current view of hfoff.


	MTM
	
	TC has been informed and told us to check the earth current, to continue if OK but be careful with next one.

	Thursday _31_A

	TBE2
	MB3080:On instructions from Pierre to analyze PTE with new AQA version, warning observed on Diode Info

Warning: VT_Ano_VT_Cat,  Diode R > 15 microOhms

Pierre told to generate NC in CDPT.
	MTM
	
	

	Friday _1_M


	TBD1
	SSS117: dipole corrector IJ cannot be powered.
	
	
	Richard investigated the problem

Meanwhile the TC asked to proceed with the MQ training.

Vtaps is the bottom of the Current lead are inverted. Richard managed to power the circuit up 60 A.

Inversion will be confirmed with the continuity test at warm (CQW2).

	Friday _1_M

Friday_1_A


	TBD1
	SSS117: Training quench 2. SPA do to a cryo problem

Second loss of Cryo OK due cold compressor down .
	ACR
	
	


	Friday_1_A


	TBB1
	unable to launch PT4: acquisition system doesn't arm
	AB/CO
	TEMA
	

	Friday_1_N


	TBD1
	During SSW@Cold(Field Strength: 760A), SPA occured...Checked with Cryo operators, No problem with "Cryo OK" signal
	?
	
	

	Friday_1_N


	TBC1
	leak on MRB side on the MB1214
	
	
	TC informed. He decided to replace the magnet on the bench.

	Friday_1_N


	TBA2
	Over-ranges in sector 01 in aperture 1 during LHC cycle.
	MTM
	
	TC informed. He asked Richard to check and if needed to replace the TRU. 2 last Tests will be repeated. Richard found that the problem came from the TRUTRU 5 installed instead of TRU 3.

	Saturday_2_M


	TBD1
	Survey cannot be done on D1, reflector support are missing. Taken from TBA2
	MTM
	
	

	Saturday_2_M


	TBA1
	As MMP program was launched on A2, results of HV program on A1cannot be seen. Even after closing the MMP program window of MMP was not closing.
	MTM
	TEMA
	Salem solved the problem.

	Saturday_2_A


	TBD1
	FS @ 760Amp: equipment missing message for Leica,PC,conditioner crate
	MTM
	
	Equipment ID for Leica & PC was entered manually by Salem. But he could not locate "id" for conditioner crate. He has suggested to use PT4 test (for setting data aquisition) & ramp-up the current manually. But the problem is not solved, Maryline will look after this problem on Monday. 

	Saturday_2_A


	TBE1
	During the de-ramp from 12850 A, SPA at 6500 A.
	ACR
	
	TC checked the quench file. He discussed with ACR team. Increase of one lead at the same time. He decided to proceed with the tests.

	Saturday_2_N


	TBD1
	HF crate is down. After trying to change to crate's power supply without success
	AB/PO
	
	Richard came took the chassis spare from cluster G. Now it is OK

	Sunday_3_M


	TBF2
	During MMP, motor 2 was not rotating, impossible to launch the test, we continue with the quenches. 
	MTM
	
	Richard investigated. Start position of the shaft was too close to one end stop causing this intermittent problem. Angle of the shaft axe's with respect to the motor axe modified..

	Sunday_3_M


	TBC1
	MB1212: HV MB vs ground failed at cold BP.
	ICS
	
	Richard and TC informed. Investigation through the IFS box. MB put in stand by at cold by the TC.

It was found that the desiccant bag was inside the IFS box. TC asked to repeat the test.

Active/Passive bus bar damaged. Richard advised to continue without this.Reparation foreseen Monday.

	Sunday_3_M


	TBB1
	Powering of the MB not possible. Impossible to start RMB.
	MTM
	
	Richard came and found that the inductance was not changed from SSS to dipole

	Sunday_3_N


	TBA1
	Stop at training number 2: No alarm but TC found abnormal parasitic noises in D1-D2 low gain and DVT.
	MTM
	
	TC asked to stop the test .Not to  quench and warm the magnet. Pierre will prepare a special investigation.

	Sunday_3_N


	TBF1
	Alarm on anti-cryostat.
	ICS
	
	Heating cable ( blue cable) not connected.

	Sunday_3_N


	TBB2
	One training quench on MB1211 with SSS protection configuration (only half a circuit).MIITS ~ 37
	MTM
	
	

	Monday_4_A


	TBE1
	While launching HV test at cold before powering on MB1210, magnet name could not be entered.
	AB/CO?
	HV
	Name entered manually.











